CORBA OODS1 Tutorial 1

© 2000, Uwe Rastofer UniversitätErlangen Nürnberg,Informatik4

Reproduktion oder Verwendung dieser Unterlage zu Lehrzwecken außerhalb der Universität Erlangen ist ohne Genehmigung der Autoren unzulässig

Homework Assignment #9: Library IDL (Due Jan. 11, 2000 9:30) 0 points

For allCORBAassignments you should use the latest ORBacus 4.0 ORB.Itis installed in /local/ORBacus-4.0.3/. AlthoughCORBAprograms are supposed to be easily portable between ORB implementations, I urge you to follow my advice and use ORBacus. The assignments willuse new CORBAfeatures that are not yet available in many other ORBs. ORBacus is freely available for non-commercial use. Inthe beginningofa CORBAprojectyou willhave to design the data and interfaces that willbe remotelyaccessible. Inthisassignment youhave to design a management system for the Library. (You all should be familiar with what is going on in a Library.) a) AllIDLdefinitionshave to be in a single module Library which is stored in a file named Library.idl.

b) The entry point for the system is an interface called Librarian which provides access to Library operations and maybe other Library objects.

c) The system has to offer functionality for the following tasks:

Issue a librarycard for auser Lookup abook Borrow abook Return a book Get a list of allbooks currently borrowed on acard Get the amount ofcharges for delayed return for acard

The above items initalicsare candidates forCORBAobjects or data structures. As youshouldknowfromthe Tutorialthere are differentwaystomodelthese items.You have to givea reason whyyouchose a certain alternative.Writedownyourreasons in a file calledLibrary.txt. There is no right or wrong solution to this problem.

d) Remember that operations may fail, so you will have to throw exceptions when:

Cards become invalid after some time Books are be borrowed by someone else ... (there will be others) You can check the syntactic correctness of your IDLfile by compilingit with an IDL compiler. Please use to following command to compile your IDL:

/local/ORBacus-4.0.3/bin/jidl Library.idl